Author's response to reviews

Title: Do physician outcome judgments and judgment biases contribute to inappropriate use of treatments?
Study Protocol

Authors:

Jamie C Brehaut (jbrehaut@ohri.ca)
Roy Poses (rposes@firmfound.org)
Kaveh G Shojania (kshojania@ohri.ca)
Alison Lott (alott@ohri.ca)
Malcolm Man-Son-Hing (mhing@ohri.ca)
Elise Bassin (elise_bassin@post.harvard.edu)
Jeremy Grimshaw (jgrimshaw@ohri.ca)

Version: 4  Date: 14 May 2007

Author's response to reviews:

Dear Dr. Foy,

We agreed with your suggestions and have made the requested changes. The main point that was raised was about the referencing format. We have changed the reference numbers so that they are now in brackets and before periods; we hope that the attached draft is now in the suitable format.

Please let us know if you have any other questions or concerns.

Sincerely,
Dr. Jamie Brehaut